Author's response to reviews

Title: No effects of a 12-week supervised exercise therapy program on gait in patients with mild to moderate osteoarthritis. A secondary analysis of a randomized trial.

Authors:

Ingrid Eitzen (dringrideitzen@gmail.com)
Linda Fernandes (linda.fernandes1@gmail.com)
Lars Nordsletten (lars.nordsletten@medisin.uio.no)
May Arna Risberg (mayarna.risberg@nimi.no)

Version: 4 Date: 11 January 2015

Author's response to reviews:

Dear editorial team of Journal of Negative Results in Biomedicine,

thank you for giving us the opportunity to submit a re-revised version of our manuscript

No effects of a 12-week supervised exercise therapy program on gait in patients with mild to moderate osteoarthritis. A secondary analysis of a randomized trial.

The abstract has been shortened as requested, and is now within the word limit of 350 words.

Oslo, Norway, January 11th.

Yours sincerely

Ingrid Eitzen, PT, PhD